Author’s response to reviews

Title: The use of a new miniature cryoprobe for ablation of bone tissue: In vivo assessment of the probe and application of the method to bone in a sheep model

Authors:
Dr Frank Popken (fpopken@aol.com)
Marc Land (Land167@aol.com)
Heike Eberich (ha.erberich@uni-koeln.de)
Marfalda Bosse (m.bosse@uni-koeln.de)
Dietmar-Pierre Konig (Dietmar-Pierre.Koenig@medizin.uni-koeln.de)
Peer Eysel (peer.eysel@uni-koeln.de)

Version: 3 Date: 6 Apr 2003

Dear Mrs. Veitch

We change our accepted paper following your instructions below:

- The abstract should be rewritten slightly so that the 'aim' section is replaced by a 'background' section that explains the context of your study.

- The figure numbers, titles and legends should be removed from the graphics files and instead included in your main manuscript document, following the References (ie following the References you should list, for each figure, the number, the title of the figure, and a legend which can be up to a page of text).

We separated Fig. 1 a+b to in Fig. 1a and 1b and aditional changed Fig. 8+9 to Fig.8 and Fig 9 after converting into a powerpoint doc. We hope that the paper is no sufficient for publication.

With best wishes,
Dr. F. Popken